Author's response to reviews

Title: Severe male infertility after failed ICSI treatment - a phenomenological study of men's experiences

Authors:

Marianne Johansson (marianne.ak.johansson@vgregion.se)
Anna-Lena Hellström (anna-lena.hellstrom@fhs.gu.se)
Marie Berg (marie.berg@fhs.gu.se)

Version: 4 Date: 28 January 2011

Author's response to reviews: see over
Dear Chief Editor,

Thank you for accepting our manuscript “Severe male infertility after failed ICSI treatment- a phenomenological study of men’s experiences” for publication in Reproductive Health.

We have now checked our manuscript careful and made two very small typographical changes in the text, these are coloured in yellow.

In the paragraph Methods, Study population we have included a reproductive medicine specialist with andrology competence instead of an andrology.

Yours sincerely,

Marianne Johansson